Author's response to reviews

Title: Role of metabolically active hormones in the insulin resistance associated with short-term glucocorticoid treatment

Authors:

Jeetesh V Patel (Jeetesh.patel@swbh.nhs.uk)
David E Cummings (davidec@washington.edu)
John P Girod (girodjp@upmc.edu)
Alwin V Mascarenhas (alwin_vm@yahoo.com)
Elizabeth A Hughes (elizabeth.hughes@swbh.nhs.uk)
Manjula Gupta (guptam@ccf.org)
Gregory Y H Lip (gregory.lip@swbh.nhs.uk)
Sethu Reddy (reddys@ccf.org)
Daniel J Brotman (brotman@jhmi.edu)

Version: 4 Date: 11 September 2006

Author's response to reviews:

Final changes by BMC